Status of DVCS Calorimeter

Kijun Park

Jan. 18, 2015

DVCS Collaboration meeting



Crystal inventory photo-collections

— visual inspection all six surfaces
https://hallaweb. jlab.org/experiment/DVCS/DVCS_Crystals/
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https://hallaweb.jlab.org/experiment/DVCS/DVCS_Crystals/

Calo. Crystal de-stacking

DVCS2016

@ Notes

)

Start de-stacking Nov.16
and completed Nov.18
(Thanks to Tom)

Many blocks were stuck
together due the remnant

of adhesive that used for
shim .
— even worse during the

experiment due to hot
temperature inside of
dark-box (talk about
later)

Take a note for any
specific condition (broken
brass/screw/optical
glue/chip/...)

Secure block in white box
and store them into each
container row-by-row

Survey by RadCon group

@ Nov.24 .
Completed transporting

all crystals to ODU @
Nov.25
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Tables

Physical Repairs

Cry.Num Issue Est.

103,134 broken brass 2hrs

128,228 broken poly.screw 2hrs
10,67,112,126,234,245 chip on face no need to fix

66,67,69,72,75,80,81,95,117,132,235 low performance” 11hrs

*: Mongi's list from elog#271
*: We will just replace the PMT but keep the old base (Thanks to Thomas Hartlove, Mathieu Ehrhart)

*** not include additional repair hours if inspection finds and issue Block replaced with: 271 — 144/ 200 — 206/ 205 — 205

Completed replacing PMT and will deliver blocks to the Hall on
Jan. 19 ! We can start to test some of them in this week !
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Position Survey for New dark-box

DarkBox
Calorimeter

Calorimeter
Stand

S_tail

Currently, Calorimeter is located at

angle = apprx. 16(deg)

Distance from target : 3.325(m)

K.Park

~S_head

(38”) S_wid : Width of Calo Stand
(88”) S_Len : Length of Calo Stand
(24”) D_wid : Width of Calo DakIBox
(36”) D_Len : Length of Calo DarkBox

S >
D_beaml

(9”) GL: Left side Gap btw Calo
Stand edge and DarkBox edge

(4.5”) GR : Right side Gap btw Calo
‘% Stand edge and DarkBox edge
2
<. (10.5”)S_head : Front side Gap btw Calo
2 Stand edge and DarkBox edge

(42”)S_tail : Tail side Gap btw Calo

I | » Stand edge and DarkBox edge
D_beam2

(33”) D_beam1 : Distance btw DarkBox edge
and beeam pipe edge at pointl

(34.5”) D_beam?2 : Distance btw DarkBox edge
and beeam pipe edge at point2

DVCS2016 Jan. 18, 2015
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New dark-box @ Aug. 18, 20

@ New dark-box with active ventilation system (Thanks to Thomas Hartlove, ODU)
— Two ventilate-fans (inhale/exhale),

DVCS2016 Jan. 18, 2015



Active Air Cooling System

@ Sitting position: side & back of
dark-box

@ Easy to attach/detach, polyken
tape: light tight

@ Maze pipe: light tight

@ 4way-out of Dampered roof cap:
light tight

@ Model: DB204 (4x4-in duck
fan)

@ 65 CFM free air (80 CFM as
Max)

@ Thermally protected Class B
motor

@ Weight: 1.5LBS

DVCS2016 Jan. 18, 2015 7 /10



Temperature

— lha — DVCS dork box
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@ Starting Temp=100°F (when LV/HV were off) with OLD dark-box
@ Temperature monitoring with e-beam during 2014 commissioning run

@ Deviation of temp is about 28°F (16°C) as maximum between on and off the
HV/LV/Beam
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Temperature Monitoring with new dark-box @ Sep., 2015

write.temp.dvcs.log write.temp.dvcs.log

T I T T T T T T T T
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@ Starting Temp=100°F (when LV/HV were off), see how much Temp. change

@ Take a cosmic run (Run number= 11556)

1/ Engaged LV/HV as cosmic mode

2/ Temp. monitoring is running as cron job on dvesdaq2

3/ Run overnight (24hrs)

Deviation of temp is about 7°F (4°C) as maximum between on and off the HV/LV

@ Fan creates a little vibration (I can feel it by hand on the box), Check any impact on
signal/pedestal
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Pedestal Monitoring with new dark-box @ Sep.28, 2015

@ Monitoring ARS after taking cosmic runs (Run 11573(fan-off;red) vs. 11584(fan-on;blue))

@ Little electronic noise effect in pedestal & Link in detail comparison
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http://hallaweb.jlab.org/experiment/DVCS/DVCS_Ped_ActiveFan/

